Author's response to reviews

Title: Identification of SERPINA1 as single marker for papillary thyroid carcinoma through microarray meta analysis and quantification of its discriminatory power in independent validation

Authors:

Klemens Vierlinger (klemens.vierlinger@ait.ac.at)
Markus H Mansfeld (Markus.mansfeld@ait.ac.at)
Oskar Koperek (oskar.koperek@meduniwien.ac.at)
Christa Noehammer (christa.noehammer@ait.ac.at)
Klaus Kaserer (klaus.kaserer@meduniwien.ac.at)
Friedrich Leisch (Friedrich.Leisch@boku.ac.at)

Version: 5 Date: 29 March 2011

Author's response to reviews: see over
To The Editor

BMC Medical Genomics

Dear Editor,

thank you again for accepting our paper. The following changes were made at this stage of the revision:
- The order of the references was corrected.
- The vertical lines in the tables were removed
- Some typos were corrected

Sincerely,

Klemens Vierlinger